

Geophysical Research Letters



10.1029/2018GL077837

Key Points:

- Blocking strongly impacts both heat waves and cold spells in Europe depending on region and season
- Blocking-extreme temperature link is found realistic in the model despite its underestimation of blocking
- Blocking will continue to play a crucial role for extreme temperatures also in the future

Supporting Information:

Supporting Information S1

Correspondence to:

L. Brunner, lukas.brunner@env.ethz.ch

Citation:

Brunner, L., Schaller, N., Anstey, J., Sillmann, J., & Steiner, A. K. (2018). Dependence of present and future European temperature extremes on the location of atmospheric blocking. *Geophysical Research Letters, 45*. https://doi.org/10.1029/2018GL077837

Received 1 DEC 2017 Accepted 31 MAY 2018 Accepted article online 11 JUN 2018

©2018. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Dependence of Present and Future European Temperature Extremes on the Location of Atmospheric Blocking

Lukas Brunner^{1,2,3}, Nathalie Schaller⁴, James Anstey⁵, Jana Sillmann⁴, and Andrea K. Steiner^{1,2,6}

¹Wegener Center for Climate and Global Change, University of Graz, Graz, Austria, ²FWF-DK Climate Change, University of Graz, Graz, Austria, ³Now at Institute for Atmospheric and Climate Science, ETH Zurich, Zurich, Switzerland, ⁴Center for International Climate Research (CICERO), Oslo, Norway, ⁵Canadian Centre for Climate Modelling and Analysis, Environment and Climate Change Canada, University of Victoria, Victoria, British Columbia, Canada, ⁶Institute for Geophysics, Astrophysics, and Meteorology, Institute of Physics, University of Graz, Graz, Austria

Abstract The impact of atmospheric blocking on European heat waves (HWs) and cold spells (CSs) is investigated for present and future conditions . A 50-member ensemble of the second generation Canadian Earth System Model is used to quantify the role of internal variability in the response to blocking. We find that the present blocking-extreme temperature link is well represented compared to ERA-Interim, despite a significant underestimation of blocking frequency in most ensemble members. Our results show a strong correlation of blocking with northern European HWs in summer, spring, and fall. However, we also find a strong anticorrelation between blocking and HW occurrence in southern Europe in all seasons. Blocking increases the CS frequency particularly in southern Europe in fall, winter, and spring but reduces it in summer. For the future we find that blocking will continue to play an important role in the development of both CSs and HWs in all seasons.

Plain Language Summary Atmospheric blocking describes strong and stationary high-pressure systems at midlatitudes. It is frequently connected to surface extremes such as cold spells (CSs) and heat waves (HWs). We use a global climate model (second generation Canadian Earth System Model) with multiple realizations and investigate its performance in representing blocking and the impact on temperature extremes compared to the ERA-Interim reanalysis for present-day (1979–2010) conditions. We also look into the development in the future (2070–2099) under a high-emission scenario (Representative Concentration Pathway 8.5). We find the link between blocking and temperature extremes in Europe well represented in the second generation Canadian Earth System Model compared to ERA-Interim, despite a significant underestimation of blocking occurrence in almost all realizations. Our results show a strong impact of blocking on HWs in northern Europe we find a strong anticorrelation between blocking and HW occurrence in all seasons. Blocking increases the CS frequency in all of Europe in fall, winter, and spring but reduces CSs in summer. For the future we find that blocking will continue to play an important role in the development of both CSs and HWs in all seasons.

1. Introduction

There is ever increasing evidence that anthropogenic climate change affects society through an increase in extreme weather events and will continue to do so in the future (IPCC, 2013; Fischer & Knutti, 2015). Understanding the physical mechanisms behind high-impact extremes such as heat waves (HWs) or cold spells (CSs) is therefore essential for policy makers in order to be able to take informed decisions (Zhang, 2013). One of the main drivers of European weather and climate variability is the frequent occurrence of atmospheric blocking (Woollings, 2010). Blocking describes strong and stationary high-pressure systems, which interrupt the climatological flow at midlatitudes for up to several weeks (Rex, 1950). They are often linked to the development of surface HWs (e.g., Pfahl & Wernli, 2012) due to their persistence, which allows the accumulation of heat in a certain area (Bieli et al., 2015; Perkins, 2015). The Russian HW in summer 2010 is one recent example of a block leading to a devastating HW, which affected large parts of Europe and Russia. The exceptionally hot temperatures, which lasted for several weeks, led to over 50,000 additional deaths and billions of US Dollars



ഹ

in economic losses in Russia alone (e.g., Barriopedro et al., 2011). The role of the extraordinary atmospheric conditions caused by the blocking in the development and maintenance of the Russian HW was discussed by many studies in the aftermath (e.g., Galarneau et al., 2012; Lupo et al., 2012; Miralles et al., 2014; Schneidereit et al., 2012). Conversely, blocking has also been connected to winter CSs (e.g., Buehler et al., 2011; Sillmann et al., 2011), mainly through the advection of cold air (e.g., Bieli et al., 2015; Sousa et al., 2018). Recently, also the blocking impacts during the transition seasons such as spring have come into focus (e.g., Brunner et al., 2017; Cassou & Cattiaux, 2016). Extreme temperatures in spring can have strong impacts on vegetation, as plants are particularly vulnerable during the early green-up phase (e.g., Hufkens et al., 2012). In spring 2016, for example, a blocking event over the British Isles led to a CS in central and eastern Europe, causing severe damages to agricultural yields in several countries with harvest failures of up to 80% (AGRI4CAST, 2016).

Reliable knowledge on the impacts of blocking in a changing climate is therefore of utmost importance for our understanding of the development of both HWs and CSs (IPCC, 2013). However, some of the processes involved in the blocking development and maintenance are still not well understood and trends in the future evolution of blocking remain uncertain (IPCC, 2013; Woollings, 2010). In the last decades many studies have noted the limited model performance in realistically representing blocking (e.g., Anstey et al., 2013; D'Andrea et al., 1998; Davini & D'Andrea, 2016; Schiemann et al., 2017). Davini and D'Andrea (2016) find that several models have significantly improved their representation of blocking in recent years. However, the blocking frequency in the Euro-Atlantic sector is still underestimated by as much as 50% in the most recent Coupled Model Intercomparison Project Phase 5 (CMIP5) models (Davini & D'Andrea, 2016). For the future a decrease in blocking frequency and magnitude in the Euro-Atlantic sector has been predicted, depending on the season (Kennedy et al., 2016; Matsueda & Endo, 2017). These considerations raise the question of the representation of the link between blocking and extreme temperature occurrence in models. Masato et al. (2014) investigate the impact of winter blocking on European temperature extremes during present and future conditions. They find that the models included in their study represent the relationship between blocking and temperature extremes well, given the negative bias in the blocking frequency. However, many blocking impact studies in the past have been based on only a few simulations and have focused on either the winter or summer season (e.g., Kennedy et al., 2016; Masato et al., 2014).

In this study we investigate all seasons and particularly also consider spring and fall. We use a large-ensemble approach that provides enough cases to draw from in order to allow an estimation of uncertainty due to internal variability, given that blocking is a rare phenomenon. Our study looks into the relationship between both extreme cold and extreme warm temperatures and blocking in a 50-member ensemble of the the second generation Canadian Earth System Model (CanESM2) (e.g., Fyfe et al., 2017). Our work has two main aims: (i) We evaluate the performance of the CanESM2 ensemble in representing present blocking and its link to temperature extremes compared to the ERA-Interim reanalysis (Dee et al., 2011). (ii) We investigate how present and future temperature extremes in northern and southern Europe depend on the region of the blocking and on the season.

This work is structured as follows. Section 2 describes data sets and methods used in this study. Blocking frequencies in present and future conditions are addressed in section 3, and blocking impacts on European temperature extremes are presented in section 4. A concluding discussion is given in section 5.

2. Data and Methods

2.1. Data Sets

As reference data set in this study we use the ERA-Interim reanalysis (Dee et al., 2011) at 0.75° × 0.75° spatial resolution and for the period 1981–2010. To assess the representation of the link between blocking and extreme temperatures as well as potential changes in the future, we consider a 50-member initial-condition ensemble from the CanESM2 model (Canadian Sea Ice and Snow Evolution project). The model ensemble data are available from 1950 until 2100, and we use the period 1981–2010 as present and 2070–2099 as future. The present and future periods of the simulations are driven by the CMIP5 historical and Representative Concentration Pathway 8.5 forcing scenarios, respectively. The 50 ensemble members differ only in their initial conditions; the forcing and model configuration are identical for each ensemble member. CanESM2 is a coupled atmosphere-ocean climate model that includes sea ice, land, and carbon cycle components. The atmospheric model is horizontally spectral with T63 resolution, corresponding roughly to a grid spacing of 2.5° × 2.5°. For further information on the CanESM2 model and the construction of the 50-member large ensemble, see Arora et al. (2011) and Fyfe et al. (2017).

2.2. Blocking Index

We use a standard two-dimensional blocking detection algorithm based on the reversal of 500-hPa geopotential height gradients (Anstey et al., 2013; Rex, 1950; Scherrer et al., 2006; Tibaldi & Molteni, 1990). At each grid point geopotential height gradients to the north (ΔZ_N) and to the south (ΔZ_S) are calculated as

$$\Delta Z_{\rm N} = \frac{Z(\lambda, \phi + \Delta\phi) - Z(\lambda, \phi)}{\Delta\phi}$$

$$\Delta Z_{\rm S} = \frac{Z(\lambda, \phi) - Z(\lambda, \phi - \Delta\phi)}{\Delta\phi},$$
 (1)

where Z is the geopotential height at 500 hPa, $\Delta \phi = 15^{\circ}$, and ϕ ranges from 50°N to 75°N. Instantaneous blocking (IB) is defined at a certain grid point if the gradients simultaneously fulfill $Z_N < -10m/(^{\circ}$ latitude) and $Z_S > 0m/(^{\circ}$ latitude). Spatiotemporal filtering following Woollings et al. (2008) is then applied to the IB field to select only large-scale and slow-moving cases. The maximum of the IB index over $\pm 4^{\circ}$ latitude is taken to allow for some meridional movement. Then *extended* IB cases are selected if they extend over at least 15° longitude to filter out too small systems. Finally, blocking is defined if extended IB is found within $\pm 10^{\circ}$ longitude for at least five consecutive days in order to detect only persistent and stationary systems.

Blocked days are further defined for three 30° longitude regions, referred to as the Greenland (60°W to 30°W), North Atlantic (30°W to 0°), and Scandinavian (0° to 30°E) regions. A day is considered as blocked in a given region if a block extends over more than half of the region (i.e., over more than 15° of longitude inside the region).

2.3. Heat Waves and Cold Spells

Two types of temperature extremes are considered on a daily basis for European land grid points. A HW is defined as a period of at least three consecutive days with daily maximum temperature above the maximum temperature threshold at a certain grid point. This threshold is defined for each day as the 90th percentile of the climatological distribution of daily maximum temperature at the grid point in one of the two periods during a \pm 15-day window centered on the calendar day. Equivalently, a CS is defined as a period of at least three consecutive days with daily minimum temperature below the minimum temperature threshold. Analogously to the HW case, the minimum temperature threshold is defined as the 10th percentile of climatological minimum daily temperature (e.g., Russo et al., 2015).

Climatological HW/CS frequencies as well as the HW/CS frequencies during blocked days are calculated on a grid point basis for the two 30-year periods representing present and future conditions. For both periods the relative anomaly is then computed for each grid point following:

$$x_{\text{anom}} = \frac{x_{\text{blocking}}}{x_{\text{clim}}},\tag{2}$$

where x_{blocking} is the HW/CS frequency during blocked days and x_{clim} is the climatological frequency during all days (including blocked days).

We also calculate the mean HW/CS anomaly on the land surface in two regions: northern Europe (15° W to 30° E and 50° N to 75° N), and southern Europe (15° W to 30° E and 35° N to 50° N). Since we show the anomalies as multiples of the climatological frequency, single outlier grid cells (where $x_{blocking}$ is very small) can have potentially large impacts on the area mean in this view. We therefore restrict the anomalies to decreases by one fifth and increases by a factor 5 for the calculation of the area mean.

2.4. Significance Testing

To establish the statistical significance of the link between blocking and HWs/CSs, we use a Monte Carlo test. For a certain number of blocked days *N* in a given setting (i.e., region, period, season, and ensemble member) the same number of *N* random days is drawn. To conserve autocorrelation, consecutive blocked days are clustered and lead to clusters of the same size in the random samples. The random draw is repeated 100 times, and from this distribution we use the 5th and 95th percentiles as thresholds for statistical significance. A certain value can therefore be either statistically significantly higher or lower, or not significantly different. In the figures showing the ensemble mean, the percentage of ensemble members showing statistical significance is indicated. Only values with no contradiction (i.e., one member showing a significantly higher and another member a significantly lower value) in the significance measure are indicated.





Figure 1. Percentage of blocked days per region and season for ERA-Interim (× signs) and the CanESM2 ensemble mean (colored lines) and ensemble standard deviation (colored boxes) during present (blue) and future (red) conditions (see Table S1 in the supporting information for underlying statistics). CanESM2 = second generation Canadian Earth System Model; DJF = December–February; MAM = March–May; JJA = June–August; SON = September–November.

3. Blocking Frequencies in Present and Future

We first investigate the present-day blocking representation in the CanESM2 ensemble members and compare it to ERA-Interim. Figure 1 shows the relative number of blocked days in all regions and seasons for ERA-Interim and the CanESM2 ensemble mean and standard deviation. For most seasons and regions the number of blocked days is underestimated by the model mean by up to a factor 2 or more compared to ERA-Interim. Only for the North Atlantic and Scandinavia regions in spring the ERA-Interim blocking frequency lies within one standard deviation of the CanESM2 ensemble mean. For the future, we find a strong decrease in the number of blocked days across all seasons and regions in agreement with recent literature (e.g., Kennedy et al., 2016; Matsueda & Endo, 2017). However, we notably do not find any significant change in the Scandinavian region, which is in line with Masato et al. (2014), who note that the decrease of blocking occurrence in the Euro-Atlantic region strongly depends on the selection of the region and may rather be a shift to the east. Indeed, additionally investigating the region from 30°E to 60°E reveals a slight future increase in the blocking frequency in summer and fall.

4. Blocking Impacts

4.1. Blocking Impacts on Present Winter Cold Spells

The link between blocking in the three different regions and the CS frequency in present-day European winters (December–February) is shown

in Figure 2. Despite the considerable discrepancy in the blocking representation discussed above, the extreme temperature response to blocking in all three regions is remarkably similar in the CanESM2 ensemble mean compared to ERA-Interim, in line with, for example, findings of Masato et al. (2014). Blocking over Greenland leads to a strong and statistically significant increase in the CS frequency across most of Europe, which is consistent between CanESM2 and ERA-Interim (Figures 2a and 2b). For blocking in the North Atlantic and Scandinavian regions the anomalies shift toward central Europe, while the influence in Scandinavia decreases (Figures 2c-2f). Particularly during blocking in the Scandinavian region, it is noteworthy that most of the 50 ensemble members show a more than threefold, statistically significant increase in the CS frequency in central Europe, in agreement with ERA-Interim (Figures 2e and 2f). This strong cooling effect of blocking in winter has been found to be dominantly driven by the advection of cold air (Bieli et al., 2015; Pfahl, 2014; Sousa et al., 2018). Bieli et al. (2015) calculated backward trajectories to evaluate the source region of air masses connected to cold extremes. For example, for central Europe they found that the regions of origin are mainly located to the north and east, which is consistent with advection due to blocking anticyclones. Conversely, Sousa et al. (2018) showed that for blocking located over the North Atlantic and Scandinavian regions advection is driving the negative temperature response in large parts of Europe in winter.

Considering the total number of CSs connected to blocking in CanESM2 (similar to the work of Pfahl and Wernli (2012) for summer but without the explicit colocation criterion), we find that up to 30%, 40%, and 50% of CSs coincide with blocked days in the Greenland, North Atlantic, and Scandinavian regions, respectively. Note, however, that the blocking regions are not exclusive, and a block may occur in more than one region simultaneously. Therefore, up to 70% of winter CSs in central Europe coincide with a blocking anywhere between 60°W and 30°E.

4.2. Blocking Impacts on Present Summer Heat Waves

Figure 3 shows the impact of blocking in the three regions on the HW frequency throughout Europe during present-day summers (June–August). For blocking in the Greenland region no significant signal in the HW frequency is found in CanESM2 (Figure 3a). ERA-Interim shows even weaker anomalies throughout most of Europe for blocking over Greenland. There is, however, also a distinct region of statistically significantly decreased HW frequencies in eastern Scandinavia (Figure 3b). Blocking in the North Atlantic region is correlated to HWs on the British Isles and Scandinavia, increasing the HW frequency by about a factor 2 for CanESM2 and ERA-Interim (Figures 3c and 3d). On the Iberian Peninsula, an anticorrelation is found with blocking over



CS frequency anomaly in present winters

Figure 2. Cold spell frequency anomaly during blocking in different regions (black box) for winters (December–February) in the period 1981 to 2010 for the CanESM2 ensemble mean (a, c, e) and ERA-Interim (b, d, f). Statistical significance at the 10% (two-sided) level is indicated by dots. The larger the dot size, the larger is the number of ensemble members that show statistical significance for CanESM2. Gr = Greenland; NA= North Atlantic; Sc = Scandinavia; CanESM2 = second generation Canadian Earth System Model.

the North Atlantic with HW frequencies reduced by up to one third. These findings for blocking over the North Atlantic are in good agreement with, for example, Sousa et al. (2018) who find positive temperature anomalies on the British Isles and southeastern Scandinavia as well as negative temperature anomalies on the Iberian Peninsula during blocking in this region.

A very distinct split in the HW response is visible between northern and southern Europe for blocking over Scandinavia. The HW occurrence in Scandinavia is increased by more than a factor of 3 and statistically significant across all ensemble members, during blocking over Scandinavia. In sharp contrast, southern Europe shows a decrease in HW frequency by a factor of 3, however, with fewer ensemble members showing statistical significance. The response pattern shows excellent agreement between CanESM2 (Figure 3e) and ERA-Interim (Figure 3f), except for the Iberian peninsula, where about 50% of CanESM2 ensemble members show a statistically significant decrease in the HW frequency, while ERA-Interim shows hardly any signal. This split can be attributed to different processes driving the temperature response. As has been shown by Sousa et al. (2018) for summer blocking over Scandinavia, diabatic processes play a dominant role at the location of the block and there are strong colocated positive radiation anomalies during blocking over this region. Since we define blocking as stationary high pressure at midlatitudes following a classical definition (see section 2.2), we only consider cases northward of 50°N. Blocking can hence not have the above mentioned direct effect for southern Europe (the effect of low-latitude atmospheric ridges on southern Europe is addressed in Sousa et al. (2018)). Therefore, advection of cooler air from the north due to the anticyclonic motion of the block is most probably again the main driver of the strongly reduced HW frequency in southern Europe similar to the winter CS cases.

In total more than 60% of Scandinavian summer HWs are connected to blocking over Scandinavia in the CanESM2 ensemble mean. For ERA-Interim the value reaches 80%, consistent with findings of Pfahl and Wernli (2012). At the same time, less than 10% of HWs in southern Europe co-occur with blocking



HW frequency anomaly in present summers

Figure 3. Same as Figure 2 but for summer (June–August) heat waves. The dashed boxes in (f) indicate northern Europe (NE) and southern Europe (SE) regions used in section 4.3.

in the Scandinavian region, highlighting again the anticorrelation between blocking and HWs in some parts of Europe also in summer.

4.3. Present and Future European Extreme Temperature Response to Blocking Across Seasons

In the following we consider area averages over northern and southern Europe (as indicated in Figure 3f) and investigate ensemble mean and spread (see section 2.3 for details). This approach is motivated by the distinct split in the HW response to blocking over Scandinavia in present-day summers (Figures 3e and 3f), which is also found in future summers (Figure S9) as well as in present and future spring (Figures S3 and S8), fall (Figures S6 and S10), and to some extent in winter (Figures S1 and S7). A split can also be found for the CS response to blocking over Scandinavia, particularly for ERA-Interim and present-day conditions in CanESM2 (Figures 2, S2, S4, and S5).

Figure 4 provides a summary on the dependence of the European CS and HW frequency on the region of atmospheric blocking during present-day and future conditions. In the following we discuss all seasons, addressing (i) the CanESM2 performance compared to ERA-Interim, and (ii) the temperature response to blocking in different regions for present and future conditions. Corresponding maps can be found in the supporting information (Figures S1–S10).

Regarding the impact of winter blocking on CSs, CanESM2 and ERA-Interim are in good agreement (Figures 4a and 4b). Larger deviations are found for HWs, with a general underestimation in CanESM2. The difference in the sign of the HW anomaly between CanESM2 and ERA-Interim for southern Europe during blocking over Greenland is based only on nonsignificant anomalies (Figure S1). The strong negative HW anomaly in northern and southern Europe during blocking over Scandinavia seen in ERA-Interim is captured by none of the 50 CanESM2 ensemble members, while the spatial distribution compares well between reanalysis and model.

Hardly any changes are apparent in the winter CS and HW response to blocking in the future compared to the present, so that blocking can be expected to play the same important role in the development of winter cold extremes also in a warmer climate.



Mean CS/HW anomaly per season in northern and southern Europe



For spring CSs the ensemble interquartile range is notably narrow indicating excellent agreement on the blocking-CS link across most ensemble members (Figures 4c and 4d). The response is also in good agreement with ERA-Interim for present-day conditions, except for North Atlantic blocking and northern Europe CSs, where ERA-Interim shows a decrease in CSs for the northernmost part of Scandinavia (Figure S2).

Regarding HWs, good agreement in northern Europe is found, while in southern Europe HWs are underestimated by most CanESM2 ensemble members. Present and future spring HWs show a particularly strong sensitivity to the blocking region, similar to summer (Figure 3). Blocking over Greenland leads to a decreased HW frequency with a large ensemble spread. Blocking over the North Atlantic and Scandinavia has an opposing effect on northern and southern Europe: HWs in northern Europe increase during blocking by a factor of 2 in the area mean, while HWs in southern Europe decrease to one third. This strong sensitivity of the temperature response to the exact location of the block highlights the importance of further looking into atmospheric dynamics in spring as has been done, for example, by Cassou and Cattiaux (2016).

In present summers, both CS and HW anomalies in CanESM2 are in good agreement with ERA-Interim (Figures 4e and 4f). However, during blocking over Greenland they show a large spread across ensemble members, with the ensemble means indicating a tendency toward lower temperatures. As discussed in section 4.2, blocking over the North Atlantic and Scandinavia has the opposite effect on northern and southern Europe. This effect is strongest during blocking over Scandinavia, increasing the HW frequency in northern Europe almost by a factor of 2.5, while at the same time decreasing it in southern Europe by a factor of 2.5 during present day as well as future conditions (Figures 3 and S9).

For fall blocking (Figures 4g and 4h), the CS response is comparable to winter (Figure 4a), while the HW response is comparable to summer (Figure 4f). In particular, the split HW response for northern and southern Europe is also apparent in fall. Furthermore, fall is the only season where some of the future HW responses are different from present conditions in a significant number of ensemble members. It is, however, noteworthy that hardly any of the (present or future) responses to blocking over Greenland and the North Atlantic are statistically significant in the spatially resolved view (Figures S6 and S10).

5. Discussion and Conclusions

We used a large, 50-member ensemble of the CanESM2 model to investigate the impact of blocking high-pressure systems in different regions and seasons on temperature extremes in Europe during present and future conditions.

For present-day conditions (1981–2010), the link between blocking and extreme temperatures is well represented despite an underestimation of blocking occurrence in CanESM2 compared to ERA-Interim for most seasons and regions. For the future (2070–2099), the ensemble mean shows a robust decrease in the number of blocked days over Greenland and the North Atlantic, while no significant trend is apparent over Scandinavia. These findings are in agreement with recent research on the evolution of blocking under climate change (e.g., Kennedy et al., 2016; Matsueda & Endo, 2017).

We find blocking to be linked to warm conditions in summer and — to a certain degree — also in spring and fall, with blocking significantly increasing the HW frequency in northern Europe during those seasons. This is in agreement with earlier studies connecting blocking to an increase in colocated summer HWs, which is found to be mainly driven by increased radiative forcing at the location of the block (e.g., Meehl & Tebaldi, 2004; Pfahl & Wernli, 2012; Sousa et al., 2018). However, our results also show anticorrelation of blocking with HWs in southern Europe for most seasons and regions. This dual impact of blocking on European HWs is consistent with a combination of increased radiative heating in the region of the block and favored cold advection on the eastern and southern flanks as discussed in section 4.2. In this regard our findings also show that the response of northern European HWs to blocking highly depends on the region of the block in spring, summer, and fall. Blocking west of Europe over Greenland tends to lower the HW frequency in northern Europe, while blocking further east over Scandinavia increases it.

The blocking impact on CSs is found to be strongest in winter, consistent with the literature (e.g., Buehler et al., 2011; Sillmann et al., 2011). Blocking anywhere in the Euro-Atlantic region increases the CS frequency in all of Europe in winter and mostly also in spring and fall, mainly driven by cold advection (see section 4.1). In summer, blocking conversely leads to a decrease in the occurrence of CSs as radiative effects become more important as also noted by Sousa et al. (2018).

Despite the general underestimation of blocking in CanESM2, the link between present-day blocking and HW and CS occurrences in Europe is found to be well represented compared to ERA-Interim in most of the investigated regions, seasons, and ensemble members. This is in agreement with findings by Masato et al. (2014) and gives confidence in model-based impact studies, which, for example, assess associated risks (Zscheischler & Seneviratne, 2017). For the future we find that the blocking link to extreme temperatures is essentially preserved, indicating that blocking will continue to play an essential role in the development

Acknowledgments

The authors thank S. Russo (EC DK Joint Research Centre, Italy) for providing the heat wave index routine. ECMWF (Reading, UK) is acknowledged for access to ERA-Interim and the Canadian Centre for Climate Modelling and Analysis for making available the CanESM2 simulations. This work was funded by the Austrian Science Fund (FWF) under research grant W 1256-G15 (Doctoral Programme Climate Change—Uncertainties, Thresholds and Coping Strategies). L. Brunner was financially supported by a Marietta Blau grant by the Austrian Exchange Service (OeAD), financed by funds from the Austrian Federal Ministry of Science, Research and Economy (BMWFW). J. Sillmann and N. Schaller were supported by ClimateXL (project 243953) funded through the Norwegian Research Council.

of CSs and HWs (e.g., Schaller et al., 2018). Our work therefore shows that continued research to increase the understanding of the physical mechanisms behind the link between blocking and both HWs and CSs, not only in summer and winter but in all seasons, is essential for predicting future climate impacts and associated risks.

References

AGRI4CAST (2016). JRC MARS Bulletin—Crop monitoring in Europe.

- Anstey, J. A., Davini, P., Gray, L. J., Woollings, T. J., Butchart, N., Cagnazzo, C., et al. (2013). Multi-model analysis of Northern Hemisphere winter blocking: Model biases and the role of resolution. *Journal of Geophysical Research: Atmospheres*, 118, 3956–3971. https://doi.org/10.1002/jgrd.50231
- Arora, V. K., Scinocca, J. F., Boer, G. J., Christian, J. R., Denman, K. L., Flato, G. M., et al. (2011). Carbon emission limits required to satisfy future Representative Concentration Pathways of greenhouse gases. *Geophysical Research Letters*, 38, L05805. https://doi.org/10.1029/2010GL046270
- Barriopedro, D., Fischer, E. M., Luterbacher, J., Trigo, R. M., & García-Herrera, R. (2011). The hot summer of 2010: Redrawing the temperature record map of Europe. *Science*, 332(6026), 220–224. https://doi.org/10.1126/science.1201224
- Bieli, M., Pfahl, S., & Wernli, H. (2015). A Lagrangian investigation of hot and cold temperature extremes in Europe. *Quarterly Journal of the Royal Meteorological Society*, 141(686), 98–108. https://doi.org/10.1002/qj.2339

Brunner, L., Hegerl, G. C., & Steiner, A. K. (2017). Connecting atmospheric blocking to European temperature extremes in spring. *Journal of Climate*, 30(2), 585–594. https://doi.org/10.1175/JCLI-D-16-0518.1

- Buehler, T., Raible, C. C., & Stocker, T. F. (2011). The relationship of winter season North Atlantic blocking frequencies to extreme cold or dry spells in the ERA-40. *Tellus A*, 63(2), 212–222. https://doi.org/10.1111/j. 1600-0870.2010.00492.x
- Cassou, C., & Cattiaux, J. (2016). Disruption of the European climate seasonal clock in a warming world. *Nature Climate Change*, *6*, 589–594. https://doi.org/10.1038/nclimate2969
- D'Andrea, F., Tibaldi, S., Blackburn, M., Boer, G., Déqué, M., Dix, M. R., et al. (1998). Northern hemisphere atmospheric blocking as simulated by 15 atmospheric general circulation models in the period 1979–1988. *Climate Dynamics*, *14*(6), 385–407. https://doi.org/10.1007/s003820050230
- Davini, P., & D'Andrea, F. (2016). Northern hemisphere atmospheric blocking representation in global climate models: Twenty years of improvements? *Journal of Climate*, 29(24), 8823–8840. https://doi.org/10.1175/JCLI-D-16-0242.1
- Dee, D. P., Uppala, S. M., Simmons, A. J., Berrisford, P., Poli, P., Kobayashi, S., et al. (2011). The ERA-Interim reanalysis: Configuration and performance of the data assimilation system. *Quarterly Journal of the Royal Meteorological Society*, 137(656), 553–597. https://doi.org/10.1002/qj.828
- Fischer, E. M., & Knutti, R. (2015). Anthropogenic contribution to global occurrence of heavy-precipitation and high-temperature extremes. *Nature Climate Change*, 5, 560–564. https://doi.org/10.1038/nclimate2617
- Fyfe, J. C., Derksen, C., Mudryk, L., Flato, G. M., Santer, B. D., Swart, N. C., et al. (2017). Large near-term projected snowpack loss over the western United States. *Nature Communications*, 8, 14996. https://doi.org/10.1038/ncomms14996
- Galarneau, T. J., Jr, Hamill, T. M., Dole, R. M., & Perlwitz, J. (2012). A multiscale analysis of the extreme weather events over western Russia and northern Pakistan during July 2010. *Monthly Weather Review*, *140*, 1639–1664. https://doi.org/10.1175/MWR-D-11-00191.1
- Hufkens, K., Friedl, M. A., Keenan, T. F., Sonnentag, O., Bailey, A., O'Keefe, J., & Richardson, A. D. (2012). Ecological impacts of a widespread frost event following early spring leaf-out. *Global Change Biology*, *18*(7), 2365–2377. https://doi.org/10.1111/j.1365-2486.2012.02712.x
- IPCC (2013). Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. In T. F. Stocker, et al. (Eds.) (1535 pp.). Cambridge, UK: Cambridge University Press. Kennedy, D., Parker, T., Woollings, T., Harvey, B., & Shaffrey, L. (2016). The response of high-impact blocking weather systems to climate
- change. Geophysical Research Letters, 43, 7250–7258. https://doi.org/10.1002/2016GL069725
- Lupo, A. R., Mokhov, I. I., Akperov, M. G., Chernokulsky, A. V., & Athar, H. (2012). A dynamic analysis of the role of the planetary- and synoptic-scale in the summer of 2010 blocking episodes over the European part of Russia. *Advances in Meteorology*, 2012, 11. https://doi.org/10.1155/2012/584257
- Masato, G., Woollings, T., & Hoskins, B. J. (2014). Structure and impact of atmospheric blocking over the Euro-Atlantic region in present-day and future simulations. *Geophysical Research Letters*, 41, 1051–1058. https://doi.org/10.1002/2013GL058570
- Matsueda, M., & Endo, H. (2017). The robustness of future changes in northern hemisphere blocking: A large ensemble projection with multiple sea surface temperature patterns. *Geophysical Research Letters*, 44, 5158–5166. https://doi.org/10.1002/2017GL073336
- Meehl, G. A., & Tebaldi, C. (2004). More intense, more frequent, and longer lasting heat waves in the 21st century. *Science*, 305(5686), 994–997. https://doi.org/10.1126/science.1098704
- Miralles, D. G., Teuling, A. J., van Heerwaarden, C. C., & de Arellano, J. V.-G. (2014). Mega-heatwave temperatures due to combined soil desiccation and atmospheric heat accumulation. *Nature Geoscience*, 7(5), 345–349. https://doi.org/10.1038/ngeo2141
- Perkins, S. E. (2015). A review on the scientific understanding of heatwaves Their measurement, driving mechanisms, and changes at the global scale. Atmospheric Research, 164, 242–267. https://doi.org/10.1016/j.atmosres.2015.05.014
- Pfahl, S. (2014). Characterising the relationship between weather extremes in Europe and synoptic circulation features. *Natural Hazards and Earth System Sciences*, 14(6), 1461–1475. https://doi.org/10.5194/nhess-14-1461-2014
- Pfahl, S., & Wernli, H. (2012). Quantifying the relevance of atmospheric blocking for co-located temperature extremes in the Northern Hemisphere on (sub-)daily time scales. *Geophysical Research Letters*, *39*, L12807. https://doi.org/10.1029/2012GL052261
- Rex, D. F. (1950). Blocking action in the middle troposphere and its effect upon regional climate I: An aerological study of blocking action. *Tellus*, 2(3), 196–211. https://doi.org/10.1111/j.2153-3490.1950.tb00331.x
 - Russo, S., Sillmann, J., & Fischer, E. M. (2015). Top ten European heatwaves since 1950 and their occurrence in the coming decades. Environmental Research Letters, 10(12), 124003. https://doi.org/10.1088/1748-9326/10/12/124003
 - Schaller, N., Sillmann, J., Anstey, J., Fischer, E. M., Grams, C. M., & Russo, S. (2018). Influence of blocking on Northern European and Western Russian heatwaves in large climate model ensembles. *Environmental Research Letters*, 13, 54015. https://doi.org/10.1088/1748-9326/aaba55
- Scherrer, S. C., Croci-Maspoli, M., Schwierz, C., & Appenzeller, C. (2006). Two-dimensional indices of atmospheric blocking and their statistical relationship with winter climate patterns in the Euro-Atlantic region. *International Journal of Climatology*, 26(2), 233–249. https://doi.org/10.1002/joc.1250

- Schiemann, R., Demory, M.-E., Shaffrey, L. C., Jane, S., Vidale, P. L., Mizielinski, M. S., et al. (2017). The resolution sensitivity of northern hemisphere blocking in four 25-km atmospheric global circulation models. *Journal of Climate*, 30(1), 337–358. https://doi.org/10.1175/JCLI-D-16-0100.1
- Schneidereit, A., Schubert, S., Vargin, P., Lunkeit, F., Zhu, X., Peters, D. H. W., & Fraedrich, K. (2012). Large-scale flow and the long-lasting blocking high over Russia: Summer 2010. *Monthly Weather Review*, *140*(9), 2967–2981. https://doi.org/10.1175/MWR-D-11-00249.1 Sillmann, J., Croci-Maspoli, M., Kallache, M., & Katz, R. W. (2011). Extreme cold winter temperatures in Europe under the influence of North
- Atlantic atmospheric blocking. Journal of Climate, 24, 5899–5913. https://doi.org/10.1175/2011JCLI4075.1 Sousa, P. M., Trigo, R. M., Barriopedro, D., Soares, P. M. M., & Santos, J. A. (2018). European temperature responses to blocking and ridge
- regional patterns. *Climate Dynamics*, 50(1), 457–477. https://doi.org/10.1007/s00382-017-3620-2

Tibaldi, S., & Molteni, F. (1990). On the operational predictability of blocking. *Tellus A*, 42(3), 343–365. https://doi.org/10.1034/j.1600-0870.1990.t01-2-00003.x

- Woollings, T. (2010). Dynamical influences on European climate: An uncertain future. *Philosophical Transactions of the Royal Society*, 368(1924), 3733–3756. https://doi.org/10.1098/rsta.2010.0040
- Woollings, T., Hoskins, B., Blackburn, M., & Berrisford, P. (2008). A new Rossby wave-breaking interpretation of the North Atlantic Oscillation. Journal of the Atmospheric Sciences, 65(2), 609–626. https://doi.org/10.1175/2007JAS2347.1
- Zhang, S. (2013). Technique description for the excess phase data processing for radio occultation (Tech. Rep.). SPACE Research Centre, RMIT.
- Zscheischler, J., & Seneviratne, S. I. (2017). Dependence of drivers affects risks associated with compound events. *Science Advances*, 3(6). https://doi.org/10.1126/sciadv.1700263